

Future Science OA



ISSN: (Print) (Online) Journal homepage: www.tandfonline.com/journals/ifso20

Societal Influence and Psychological Distress Among Indonesian Adults in Java on the Early Omicron Wave of COVID-19

Annette d'Arqom, Muh Akram, Nafesa Shafira Azzahranisa, Mhd Zamal Nasution, Endang Retno Surjaningrum & Junaidah Yusof

To cite this article: Annette d'Arqom, Muh Akram, Nafesa Shafira Azzahranisa, Mhd Zamal Nasution, Endang Retno Surjaningrum & Junaidah Yusof (2023) Societal Influence and Psychological Distress Among Indonesian Adults in Java on the Early Omicron Wave of COVID-19, Future Science OA, 9:10, FSO894, DOI: <u>10.2144/fsoa-2023-0104</u>

To link to this article: <u>https://doi.org/10.2144/fsoa-2023-0104</u>

9	© 2023 The Authors	+	View supplementary material 🖸
	Published online: 29 Aug 2023.		Submit your article to this journal $arsigma$
<u>.III</u>	Article views: 644	Q	View related articles $oldsymbol{C}$
CrossMark	View Crossmark data 🗹	2	Citing articles: 2 View citing articles 🛽

For reprint orders, please contact: reprints@future-science.com



Societal influence and psychological distress among Indonesian adults in Java on the early Omicron wave of COVID-19

Annette d'Arqom^{*, 1,2}, Muh Akram³, Nafesa Shafira Azzahranisa³, Mhd Zamal

Nasution⁴, Endang Retno Surjaningrum⁵ & Junaidah Yusof⁶

¹Division of Pharmacology & Therapy, Department of Anatomy, Histology, & Pharmacology, Faculty of Medicine, Universitas Airlangga, 60131, Indonesia

²Airlangga Research Group for Translational Medicine & Therapeutic, Universitas Airlangga, 60131, Indonesia

³Faculty of Medicine, Universitas Airlangga, Surabaya, 60131, Indonesia

⁴Postgraduate School, Universitas Airlangga, Surabaya, 60286, Indonesia

⁵Faculty of Psychology, Universitas Airlangga, Surabaya, 60286, Indonesia

⁶School of Human Resource Development & Psychology, Faculty of Social Sciences & Humanities, Universiti Teknologi Malaysia, Johor Bahru, 81310, Malaysia

*Author for correspondence: annette-d-a@fk.unair.ac.id

Aim: Changes during the COVID-19 pandemic might create pressure on different people, thus this study aimed to measure respondents' psychological distress during the early phase of the Omicron wave in Java Island. **Methods:** A web-based survey on societal influence and COVID-19-related psychological distress was distributed through social media in November–December 2021, and received 396 responses. **Results:** This study showed that almost 50% of respondents faced psychological distress during the early phase of the Omicron variant, especially concerning hypervigilance and avoidance. Several sociodemographic factors might contribute to the incidence of psychological distress including comorbidity, age and education. **Conclusion:** Taken together, the incidence of COVID-19-related psychological distress was still found in the early phase of the Omicron variant, especially among young adults.

Plain language summary: The COVID-19 pandemic hits every aspect of human life, including the social and psychological aspects. This study was conducted to measure these changes in Indonesian adults who stayed on Java Island during the early phase of the Omicron wave. Using an online questionnaire, 396 responses were received and analyzed. The results showed that almost half of the respondents still had symptoms of psychological distress. Several sociodemographic factors might contribute to the incidence of psychological distress including comorbidity, age, and education. Understanding these factors might bring benefits for future pandemic preparedness.

Tweetable abstract: Almost half of 396 Indonesian adults in Java faced psychological distress during early phase of the Omicron wave of the COVID-19 pandemic.

First draft submitted: 11 June 2023; Accepted for publication: 2 August 2023; Published online: 29 August 2023

Keywords: anxiety • desirability • distance • information • mental health





Graphical abstract:

The COVID-19 pandemic, the biggest pandemic of the 21st history, forced all aspects of human life to change. The exact origin of SARS-CoV-2 is not yet definitively confirmed, but it is believed to have originated in bats and possibly transmitted to humans through an intermediate animal host. One of the earliest known cases emerged in a seafood market in Wuhan, China, where wild animals were sold, suggesting a possible zoonotic origin [1]. COVID-19 primarily spreads through respiratory droplets produced when an infected person coughs, sneezes, talks or breathes. The virus can also spread by touching surfaces or objects contaminated with the virus and then touching the face, especially the eyes, nose or mouth [2]. The diagnosis of COVID-19 involves several methods, including molecular tests like quantitative real-time polymerase chain reaction (qRT-PCR) and antigen tests, as well as serological tests (antibody tests) [3]. In May 2023, the WHO decided to revoke the pandemic status after decreasing in the mortality of COVID-19 cases. For more than 3 years of the pandemic, COVID-19 infects more than 765 million people, with a fatality of 6.9 million cases [4]. As of the end of the COVID-19 pandemic, the number of COVID-19 cases in Indonesia totalled 6,048,204, with 156,371 deaths [5].

The uncertainties and fears associated with the virus outbreak, mass lockdowns and economic recession causing psychological distress and mental health problems globally. A review involving 19 studies of the general population in China, Spain, Italy, Iran, the USA, Turkey, Nepal and Denmark in 2020, found several mental health disorders, such as anxiety, depression, post-traumatic stress disorder, psychological distress and stress in various incidences [6]. In Indonesia, 20–25% of people suffered anxiety symptoms, as well as depression and stress during the first and second wave of the COVID-19 pandemic [7–9]. People are significantly affected by these mental health effects as a result of psychological stressors related to the COVID-19 pandemic because loneliness, worry and depression further impair the immune system [10]. Many people were affected directly and indirectly due to SARS-CoV-2 infection and it might not have resolved within a short period of time.

The social impacts of COVID-19 could be divided into several aspects including social distancing, social anxiety, social information, social desirability and social adaptation. As social distancing was necessary to limit the spreading of SARS-CoV-2, anxiety symptoms correlated to limited social interaction, social isolation and quarantine [11]. Social anxiety due to COVID-19 has been reported widely [12]. Interestingly, a study in 386,432 students in China found that face mask knowledge and usage reduced the incidence of anxiety [13]. Increasing demand for information from many sources, including social media might increase COVID-19 knowledge and affect individual behaviors

related to COVID-19 transmission; however, much unclear information spreading also might induce mental health problems [14]. With the prolonged pandemic, the ability to adapt became important to protect someone from transmission [15]. Additionally, social desirability might influence someone's behaviors including compliance with the government's strategies [16].

A higher impact of the pandemic was felt by the most vulnerable groups, such as youth, the elderly, lowincome individuals, people with comorbidity, etc. Moreover, with the application of social distancing, decreasing in social activity and interaction, losing family members, losing a job and other stressful events caused mental health problems [9]. A study of 381 Italian COVID-19 patients found 30.2% of the respondents experienced post-traumatic stress disorder (PTSD) and 17.3% of them experiencing depressive disorder [17]. Mental health problems such as depression, anxiety, stress and post-traumatic stress disorders also reported in various countries, including USA, Poland, China, Hongkong, Thailand, Philippines, Vietnam, etc. [18-21]. Several sociodemographic factors, such as age, sex, marital status, pre-existing diseases, employment and having family, contributed to mental health problems. A study in the USA showed that older people were more likely to adapt to changes caused by COVID-19; however, the severity and mortality of COVID-19 was higher among older people [18]. Vice versa, a study in seven Asian countries concluded that younger age was more likely to develop mental health problems [20]. Sex also contributed to mental health disorders development where female has a higher risk [19,20]. Individuals with stable income or having work during the pandemic were less likely to have anxiety, depression or stress symptoms [22]. Loneliness correlated to a higher incidence of mental health issues, thus married individuals or individuals with children showed lower odds to develop these problems [20]. Interestingly, migrant workers and students have higher odds of developing anxiety, depression or post-traumatic stress disorders [19]. People with higher education levels, believe the health professional and less exposure to disinformation were less likely to have the symptoms. In addition, contact history of COVID-19 infection, being chronically ill, and having pre-existing mood disorders contributed to the development of mental health problems during the pandemic [6,19,20,23].

Unfortunately, patients might develop sequelae or new symptoms following COVID-19 recovery. Seriously ill COVID-19 patients and unvaccinated people might get long-COVID symptoms that might involve the cardio and cerebrovascular system such as tachycardia and coagulation, an endocrine system such as Type 2 diabetes, myalgia or fatigue syndrome, neurologic systems such as neurocognitive dysfunction, etc [24]. A review involving eight studies during the first one and half years of the COVID-19 pandemic showed 11–28% depressive disorders among COVID-19 survivors 12 weeks post COVID-19 infection. This incidence was not related to COVID-19 severity [25]. Moreover, fatigue and cognitive impairment were also observed to be persistent for more than 12 weeks after the infection [26].

Even though the incidence of mental health problems varies between countries, the incidence is more likely to be higher in developing countries, due to the feeling of unpreparedness with the lack of appropriate medical facilities and equipment [21]. Vaccine hesitancy and availability also play a major role in the high mortality during the second wave of the COVID-19 pandemic [27]. At the beginning of the COVID-19 vaccination program in Indonesia, the acceptance rate is 64.8% and increased to more than 90% during the Delta wave COVID-19 pandemic [28]. The governments' strategies to reduce SARS-CoV-2 transmission, such as lockdowns, strict regulation of transportation and wearing masks, vaccination program, etc. bring benefits to reduce the morbidity and mortality rate of COVID-19 cases. Those strategies also significantly lower the prevalence of clinically significant depressive symptoms [29], as reported in a study involving more than 20,000 UK adults which concludes that COVID-19 vaccination alleviates COVID-19 related psychological distress [30]. Thus, understanding the COVID-19 psychological distress, the societal impact of the pandemic, and its determinant factors are vital for a better understanding of the vulnerability's variations during the COVID-19 outbreak, thus can be beneficial to deal with future catastrophic events, especially in a diverse country like Indonesia.

Methods

Ethics, study design & data collection

This cross-sectional study was held from November to December 2021 and approved by the Health Research Ethics Committee. The survey was conducted using a survey generator (www.surveyplanet.com) and distributed online using social media and email. A landing page was provided before entering the survey providing an explanation of the aims of the study, informed consent and respondents could agree to publish an anonymous response. The consent was given by clicking the YES option, and multiple submissions were prevented as in Checklist for Reporting Results of Internet E-Surveys CHERRIES guidelines. This study followed the Declaration of Helsinki and the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guidelines. The inclusion criteria were Indonesian over 18 years old and stayed in Java Island during the early Omicron COVID-19 pandemic. The exclusion criterion was incomplete submission. To have an accurate finding and preventing the risk of reporting false-negative or false-positive findings, sample size was calculated with 5% margin of error, 95% confidence level and 100 million for population size. Using a sample size calculator (http://www.raosoft.com/samplesize.html), the minimum sample required for this study was 385 respondents.

Survey instrument

A set of questionnaires covered respondents' basic sociodemographic information, four questions to measure depression post-COVID-19 and 13 questions of the Societal Influences Survey Questionnaire (SISQ) to determine the effects of the social impact of COVID-19. Four questions were asked to measure how respondents' experienced hypervigilance, somatic symptoms, avoidance and re-experience of COVID-19 using the disaster-related psychological screening test (DRPST) during the past month. The items were measured using a five-point Likert scale (not at all (1) to extreme (5)) [31]. Thirteen out of fifteen questions of the SISQ were used to measure social distance, social anxiety, social desirability and social information during the COVID-19 pandemic. The items were measured using a four-point Likert scale (never (1) to often (4) [32]. Two questions of SISQ on social adaptation were omitted from the questionnaire due to low reliability (<0.5). The questionnaire was translated by one medical doctor, two psychologists and one social science expert prior to face validity in 20 respondents to measure respondents' level of understanding of the questions, including language, sentence and format of the question.

Analytical procedure

Respondents were divided into two groups based on their age, namely, young adult (18 to 25 years old) and middle age adult (>25 to 55 years old). The score of the DRPST was calculated using scores ranging from 4 to 20. Higher scores indicated more severe depression. Moreover, the sum of each SISQ categorized item was calculated. Higher scores indicated the respondents maintained greater distance, obtained more informative, experienced more anxiety, and adapt more to COVID-19. The reliability value was generated as a part of the pilot test. Pilot test was conducted as it is a crucial step before the actual data collection since it allows to determine whether the instrument will work successfully on a smaller scale before using it in a larger study. The pilot test for this study shows great reliability and good internal consistency of Cronbach's Alpha value. The reliability measurement on the survey data showed a Cronbach Alpha coefficient of 0.799 for DRPST, 0.649 for Social Distance, 0.657 for Social Anxiety, 0.649 for Social Desirability and 0.937 for Social Information. The Mann-Whitney U test was used to compare mean depression and each societal influential component between young adult and middle age adult groups. To investigate the predictive factors of each symptom of depression, binary logistic regression was used. To perform this analysis, a Likert scale of DRPST was contracted to 0 = not experienced the symptom, and 1 = experienced the symptom. Significant predictive factors were defined as variables with p < 0.05.

Results

Characteristics of respondents

Three hundred ninety-six valid and completed responses were analysed. More than 70% of respondents were young adults (72.72%) and 67.42% were females. Ratio between high school graduates and higher-level graduates was almost 1 to 1. Only 20% of respondent were ever diagnosed with COVID-19 and 17.42% presented a comorbidity. Almost 77% of respondents reported that their self-claimed economic status compared with their neighbors was average. Characteristics of respondents are summarized in Table 1.

COVID-19 related psychological distress & its determinant factors

Four questions of the DRPST questionnaire screened major depressive disorder or post-traumatic stress disorder (DRPST) after a disaster. This study showed that during the early wave of the Omicron variant, no significant differences were noted of the incidence of psychological distress among young and middle age adults including hypervigilance (46.5 vs 42.6%), somatic symptoms (17.7 vs 13%). While the incidence of avoidance (51 vs 38.9%, p = 0.031) and re-experiencing (38.5 vs 27.8%, p = 0.047) were significantly higher in young adults (Figures 1A to D). However, using a 5 point-Likert-scale (1 to 5), the results showed a significant difference between mean total score from young adults 6.42 ± 2.42 and middle age adults 5.85 ± 2.37 (p = 0.009) on total score of DRPST. Significant differences between young and middle age adults were also observed in avoidance (1.85 \pm 1.01 vs

Table 1. Characteristics of respondents.							
Sociodemographic factor	Young adult (n = 288)	Middle age adult (n = 108)					
Sex							
Male	77 (26.7%)	52 (48.1%)					
Female	211 (73.3%)	56 (51.9%)					
Education							
High school graduate	208 (72.2%)	4 (3.7%)					
University graduate	80 (27.8%)	104 (96.3%)					
History of COVID							
Never diagnosed	233 (80.9%)	83 (76.9%)					
Diagnosed	55 (19.1%)	25 (23.1%)					
Comorbidity							
Does not have	250 (86.8%)	77 (71.3%)					
Have comorbidity	38 (13.2%)	31 (28.7%)					
Economy status							
Average and below	220 (76.4%)	82 (75.9%)					
Above average	68 (23.6%)	26 (24.1%)					
Supplement consumption							
Consumed	219 (76%)	86 (79.6%)					
Not consumed	69 (24%)	22 (20.4%)					



Figure 1. Incidence of COVID-19-related psychological distress among young and middle age adults in Java Island, Indonesia during early Omicron wave. (A) Hypervigilance. (B) Somatic symptom. (C) Avoidance. (D) Re-experience.

 1.54 ± 0.77 , p = 0.008). However, no significant differences were noted concerning hypervigilance (1.75 \pm 0.92 vs 1.68 ± 0.92 , p = 0.432), somatic symptoms (1.25 ± 0.59 vs 1.21 ± 0.59 , p = 0.315) and re-experience (1.57 ± 0.85 vs 1.43 ± 0.80 , p = 0.061).

Moreover, to predict sociodemographic factors associated with respondents' COVID-19 related psychological distress, binary logistic regression was performed. Figure 2 shows the results from modelling of the outcome as a function of several independent variables including age, sex, education level, history of COVID-19, comorbidity, economic status and supplement consumption. Each component of the four symptoms of DRPST was categorized into two categories (not experiencing and experiencing). The results showed that respondents without comorbidity conditions were 1.842-times more likely to have hypervigilance (95% CI 1.045 to 3.244, p = 0.035), while respondents that consumed supplements were 1.983 more likely to experience hypervigilance (95% CI 1.023 to 2.744, p = 0.040). Moreover, respondents at younger ages were more likely to have avoidance symptoms as they were 2.439-times more likely to have avoidance symptoms (95% CI 1.339 to 4.443, p = 0.004). In addition, higher education was a negative predictor for avoidance symptoms, as respondents who graduated from high school were 0.546 less likely to have this symptom compared with respondents with a diploma or bachelor's degree (95% CI 0.323 to 0.921, p = 0.023). Our results showed that sociodemographic factors were not significantly related to somatic symptoms and re-experiencing symptoms from COVID-19 related psychological distress (Figures 2A to D).

Societal influential

Using SISQ, this study showed a total score of SISQ (37.83 ± 5.73 vs 36.30 ± 5.95 , p = 0.018) and social anxiety (10.94 ± 2.56 vs 10.17 ± 2.61 , p = 0.011) significantly differed between young and middle age adults. However, this study did not find any differences concerning social distancing (10.89 ± 3.04 vs 10.44 ± 3.06 , p = 0.167), social desirability (10.71 ± 1.36 vs 10.59 ± 1.62 p = 0.933) and social information (5.29 ± 1.61 vs 5.09 ± 1.77 , p = 0.302) between two age groups. Furthermore, no significant differences were found concerning the total SISQ and each component of societal influence when analyzed based on the sociodemographic factors (Table 2). Social anxiety showed anxiety symptoms that correlated to limited social interaction, social isolation and quarantine [11].

Discussion

Indonesia is the fourth most populated country in the world with 281,216,462 people [33]. Fifty-six percent of them reside in Java Island and make it the densest populated island in Indonesia. Because the capital of Indonesia is located on this island, the infrastructures in Java are the most developed compared with other islands [34]. However, even though Java is the most developed island, poverty remains a challenge. This inequality leads to disparities in many sectors including health, between each region in Indonesia including in Java itself [35]. With this characteristic, it make sense that almost one-half of Indonesian COVID-19 cases were reported from Java Island [36].

The COVID-19 pandemic greatly impacts health, economics, social issues, education, politics and human relationships. City lockdowns or high-scale social restrictions change and limit human habits and activities [37]. Domestic violence reportedly increased during this period [38]. Increasing chores including distance learning, has been associated with depression, anxiety, and stress both within the family unit and among students [39,40]. Overload and chaotic work during the outbreak caused burnout syndrome, such as among health workers, teacher, and students, which is related to psychological distress [41–43]. Losing family members or experiencing COVID-19 itself might lead to psychological distress, anxiety, phobia, obsessive compulsive disorders, etc. [44]. For COVID-19 survivors, sequelae or a few problems might follow the recovery. Several symptoms were including persistent fatigue and cognitive impairment, which might related with neurological problems [24,26].

The incidence of depression, anxiety and stress was reported in 25% of 610 Indonesian mothers with school-age children [8]. This study showed that during the early wave of the Omicron variant, Indonesian adults in Java still faced COVID-19 related psychological distress, shown in symptoms of hypervigilance (46.5 vs 42.6%), somatic symptoms (17.7 vs 13%), avoidance (51 vs 38.9%) and re-experiencing (38.5 vs 27.8%) among young and middle age adults, respectively. Less than 50% of respondents experienced distress, that might have been caused by a high rate of vaccination coverage after the Delta wave. As of 19 November 2022, Indonesia vaccination coverage was 75.98 and 63.74%, for the first dose and second dose, respectively [45]. A study involving 910 Indonesian adults showed that younger ages, having better knowledge on vaccination, and having medical background were more likely to accept the COVID-19 vaccination program [28]. COVID-19 vaccination program also concluded to reduce distress including fear of infection, hospitalization and mortality due to COVID-19 as reported in 8090 US adults [46]. Furthermore, because Omicron occurred in the third year of the COVID-19 pandemic, the general population tended to show better resilience to the pressure compared with the early pandemic, as reviewed by Manchia *et al.* [47]. One study among 1,305 patients positive for Omicron in China also supported this finding,





Boldface numbers indicates significant p-value.

Table 2. Mann-Whitney U of societal influence.												
Sociodemographic factors	n	Social distance		Socia	Social anxiety		Social desirability		Social information		Total SISQ	
		z	p-value	z	p-value	z	p-value	z	p-value	z	p-value	
Age		-1.383	0.167	-2.545	0.011	-0.084	0.933	-1.033	0.302	-2.373	0.018	
Young adults	288											
Middle age adults	108											
Sex		-0.904	0.366	-0.982	0.326	-0.170	0.865	-0.205	0.838	-0.966	0.334	
Male	129											
Female	267											
Education		-0.887	0.375	-0.658	0.511	-0.074	0.941	-0.233	0.816	-1.009	0.313	
High school graduate	212											
University graduate	184											
History of COVID		-0.174	0.862	-0.314	0.753	-1.102	0.271	-1.590	0.112	-0.021	0.983	
Never diagnosed	316											
Ever diagnosed	80											
Comorbidity		-0.254	0.799	-0.592	0.554	-0.056	0.955	-1.239	0.215	-0.021	0.983	
Does not have	327											
Have comorbidity	69											
Economic status		-0.029	0.977	-1.576	0.115	-0.318	0.751	-0.008	0.994	-0.350	0.726	
Average and below	302											
Above average	94											
Supplement		-0.518	0.605	-0.847	0.397	-0.007	0.994	-0.228	0.820	-0.092	0.926	
Consumed	305											
Not Consumed	91											
Boldface numbers indicates significant p-value. SISO: Societal Influences Survey Ouestionnaire.												

as the prevalence of depression, anxiety and perceived stress in this population was lower than during the early pandemic (9.03, 4.60 and 17.03%, respectively) [48].

Several sociodemographic factors might be related to each distress symptom. This study found that respondents without comorbidities were 1.842-times more likely to have hypervigilance (95% CI 1.045 to 3.244, p = 0.035) or continuously assess the threat situation. Incidence of hypervigilance and depression was more likely related to COVID-19 compared with hypervigilance with anxiety, as reported globally among 7034 adults [49]. This finding contrasted with the result among 228,367 patients with COVID-19 positive test results in USA during the early pandemic, as 25.6% presented posttraumatic stress disorder (PTSD), associated with cardiovascular diseases. This study also reported that PTSD might increase the risk of hospitalization and mortality, even though the incidence was higher among patients with other psychiatric conditions other than PTSD such as depression, anxiety, alcohol disorder, bipolar disorder, etc [50].

Hypervigilance and avoidance also correlated with respondents consuming dietary supplements, as 1.983 were more likely to experience hypervigilance (95% CI 1.2 to 3.277, p = 0.008) and 1.675-times more likely to experience avoidance (95% CI 1.023 to 2.744, p = 0.040). Several nutrient deficiencies such as vitamins, minerals, amino acids, omega 3 and carbohydrates were related to mental health illness such as depression, bipolar diseases, etc. Thus, dietary supplements usually help to alleviate mental illness symptoms [51]. Moreover, a study among 84 patients with PTSD showed that 62.7% had vitamin D insufficiency and 25.4% had vitamin deficiency [52]. Interestingly, consuming micronutrients post-disaster including earthquakes, floods and massacres showed that micronutrient supplements reduced the risk of PTSD from 75 to 15% [53]. However, one study using Twitter Sentiment Analysis concluded an individuals with mental health disorders and taking a supplement showed more negative emotions in their tweets [54]. Unfortunately, this study did not measure the level of micronutrients consumed by the respondents; thus, the cause and effect between hypervigilance and consuming dietary supplements could not be concluded. However, it might be possible that respondents with hypervigilance symptoms consumed dietary supplements to reduce the risk of SARS-CoV-2. In addition, a study among 610 Indonesian mothers concluded no association

occurred between consuming supplements presenting mental health problems during the first year of the COVID-19 pandemic [8].

Moreover, respondents at younger ages were more likely to present avoidance symptoms as they were 2.439-times more likely to have avoidance symptoms (95% CI 1.339 to 4.443, p = 0.004). This study supports the finding among 111 children 3 to 8 years old and their young mothers in USA reporting 26% of the children experienced traumatic stress symptoms including hypervigilance, avoidance and intrusion as a stress response to the COVID-19 pandemic [55]. Similar findings were also reported among 610 Indonesian mothers, as younger and older mothers were more likely to develop mental health problems, compared with middle age mothers [8]. Moreover, a study among 3370 trauma-exposed adolescents from low and middle-income countries also concluded that younger age respondents exhibited a greater risk to develop post-traumatic stress events, even though they were not COVID-related events [56]. Remarkably, avoidance symptoms were usually observed to be higher among younger adults with inhibited temperament [57], which unfortunately was not investigated in this study.

In addition, higher education was a negative predictor for avoidance symptoms, as respondents graduating from high school were 0.546 less likely to have these symptoms (95% CI 0.323 to 0.921, p = 0.023) compared with respondents obtaining a diploma or higher levels. This finding differed from a study involving 2,244,193 Swedish adults concluding patients with PTSD were less likely to achieve good performance in education [58]. Additionally, a meta-analysis study involving 52 studies on PTSD and earthquakes found that respondents with lower education level exhibited a greater risk to develop PTSD [59]. This difference might have been caused because people with lower education levels did not have sufficient knowledge of COVID-19; and thus, were unafraid of or lacked stress regarding the risk. Moreover, our results showed that sociodemographic factors were not significantly related to somatic symptoms and re-experiencing symptoms from COVID-19 related psychological distress. This finding contrasted that of a study among 610 Indonesian mothers during the first year of the COVID-19 pandemic concluding no significant association existed between education level and mental health problems [8].

Age also might have contributed to societal anxiety during the early Omicron variant COVID-19 pandemic in Indonesia, as young adults were more likely to develop societal anxiety compared with middle-aged adults $(10.94 \pm 2.56 \text{ vs } 10.17 \pm 2.61, \text{ p} = 0.011)$. This might have been caused by the higher mobility of young adults compared with middle-aged adults, as during the Omicron wave, social restrictions were more relaxed. This mobility caused younger people to exhibit a greater risk of contracting SARS-CoV-2 and spreading the disease, as reported by Rumain *et al.* [60]. In addition, we could not find other sociodemographic factors as predictors for societal influence including social distancing, social desirability and social information. It might have been caused during the early Omicron period when restrictions were relaxed after significantly loosened Delta variant cases and high coverage of COVID-19 vaccination were noted [61].

During the COVID-19 outbreak, increasing in mental health disorders was unavoidable. However, the services of mental health were disrupted, such as the re-assignment of healthcare professionals, access to healthcare facilities, face mask policy might interfere with the psychiatrist examination, etc [62,63]. Thus, to eliminate this problem, internet-based cognitive behavior therapy (iCBT) was developed. It showed high effectiveness to treat depression, anxiety, mood disorders and insomnia. A combination of iCBT and psychopharmacology showed higher effectiveness to treat mental health problems compared with a single method. It was also comparable with face-to-face CBT [64–66]. A study involving 84 Hongkong university students showed a reduction of depression, anxiety and stress symptoms after completion of eight modules of iCBT [67], and the Markov model analysis concluded that iCBT was cost-saving and effective for alleviating mild anxiety [68]. Even though CBT has been used widely in Indonesia, there was no report on iCBT, especially during the COVID-19 pandemic. However, several online applications, websites, and telemedicine have been developed for screening and counseling for mental health cases [69–71].

Despite using an online questionnaire that might cause limited access to the remote respondents, instability/unfamiliarity of internet access and inability to explore the occurrence of COVID-19 burnout which might affect many individuals, our study supported the importance of controlling mental health issues after experiencing traumatic COVID-19 cases. Sociodemographic factors might necessarily be considered for more effective measurement strategies to support mental health needs during catastrophic events. Moreover, randomized sampling is required to draw complete pictures of COVID-19 related psychological distress in Java Island or the broader region.

Conclusion

The COVID-19 pandemic hits every aspect of human life, including social life and psychology. Our study shows the incidence of COVID-19-related psychological distress and social anxiety were still found in the early phase of the Omicron variant, especially among young adults. Since the psychological problems might affect the health condition and *vice versa*, understanding the COVID-19 psychological distress, the societal impact of the pandemic, and its determinant factors are vital for a better understanding of the vulnerability's variations during the COVID-19 outbreak, thus can be beneficial to deal with future catastrophic events, especially in a diverse country like Indonesia.

Summary points

- This study showed that almost 50% of respondents faced psychological distress during the early phase of the Omicron variant, especially concerning hypervigilance and avoidance.
- Several sociodemographic factors might have contributed to the incidence of psychological distress including age, comorbidities, younger age and education level.
- During the early stage of the Omicron wave, anxiety symptoms were detected, and their incidence correlated with age.
- Young adults were at risk for developing COVID-related psychological distress, which might require more attention from the stakeholders.

Author contributions

A d'Arqom designed the study, acquired the data, analyzed the data and drafted the manuscript. M Akram acquired the data, reviewed and edited the manuscript; NS Azzahranisa acquired the data, reviewed and edited the manuscript; MZ Nasution acquired the data, reviewed and edited the manuscript; J Yusof reviewed and edited the manuscript; The manuscript has been read and approved by all the authors, that the requirements for authorship as stated earlier in this document have been met, and that each author believes that the manuscript represents honest work.

Financial & competing interests disclosure

A d'Arqom received funding from LPPM Universitas Airlangga, SATU JRS, no. 1003/UN3/2022; J Yusof received funding from Universiti Teknologi Malaysia, UTM MG, PY/2022/04779 (Q.J130000.3053.04M14)/PY/2021/02555 (R.J130000.7353.4B724). The authors have no other relevant affiliations or financial involvement with any organization or entity with a financial interest in or financial conflict with the subject matter or materials discussed in the manuscript apart from those disclosed.

No writing assistance was utilized in the production of this manuscript.

Ethical conduct of research

The authors state that they have obtained appropriate institutional review board approval or have followed the principles outlined in the Declaration of Helsinki for all human experimental investigations. In addition, for investigations involving human subjects, informed consent has been obtained from the participants involved (Approved by Faculty of Medicine, Universitas Airlangga (no. 244/EC/KEPK/FKUA/2021).

Open access

This work is licensed under the Creative Commons Attribution 4.0 License. To view a copy of this license, visit http://creativecomm ons.org/licenses/by/4.0/

References

Papers of special note have been highlighted as: • of interest; •• of considerable interest

- 1. Zhou P, Yang XL, Wang XG *et al.* A pneumonia outbreak associated with a new coronavirus of probable bat origin. *Nature* 579(7798), 270–273 (2020).
- Lotfi M, Hamblin MR, Rezaei N. COVID-19: transmission, prevention, and potential therapeutic opportunities. *Clin. Chim. Acta* 508, 254–266 (2020).
- 3. Falzone L, Gattuso G, Tsatsakis A, Spandidos DA, Libra M. Current and innovative methods for the diagnosis of COVID-19 infection (Review). *Int. J. Mol. Med.* 47(6), 1–23 (2021).

- WHO. Weekly epidemiological update on COVID-19. https://www.who.int/publications/m/item/weekly-epidemiological-update-on-covid-19---4-may-2023 (2023).
- 5. Worldometer. Coronavirus Cases Indonesia. https://www.worldometers.info/coronavirus/country/indonesia/ (2023).
- Xiong J, Lipsitz O, Nasri F et al. Impact of COVID-19 pandemic on mental health in the general population: a systematic review. J. Affect. Disord. 277, 55–64 (2020).
- 7. Anindyajati G, Wiguna T, Murtani BJ *et al.* Anxiety and Its Associated Factors During the Initial Phase of the COVID-19 Pandemic in Indonesia. *Front. Psych.* 12, 634585 (2021).
- d'Arqom A, Sawitri B, Nasution Z, Lazuardi R. "Anti-COVID-19" Medications, Supplements, and Mental Health Status in Indonesian Mothers with School-Age Children. Int. J. Womens Health 13, 699–709 (2021).
- Showed 25% of the respondents had a mental health issues during the first year of COVID-19 pandemic in Indonesia.
- 9. Yusof J, d'Arqom A, Andriani AP *et al.* Dietary Supplement Consumption and Mental Health in Indonesian Adults During Second Wave of COVID-19 Pandemic. *Patient Prefer. Adher.* 17, 1799–1811 (2023).
- 10. Pourriyahi H, Yazdanpanah N, Saghazadeh A, Rezaei N. Loneliness: An Immunometabolic Syndrome. *Int. J. Environ. Res. Public Health* 18(22), 12162 (2021).
- 11. Hull HF. SARS control and psychological effects of quarantine, Toronto, Canada. Emerg. Infect. Dis. 11(2), 354-355 (2005).
- 12. Kindred R, Bates GW. The influence of the COVID-19 pandemic on social anxiety: a systematic review. Int. J.Environ. Res. Public Health 20(3), 2362 (2023).
- Xu Q, Mao Z, Wei D *et al.* Association between mask wearing and anxiety symptoms during the outbreak of COVID 19: a large survey among 386,432 junior and senior high school students in China. *J. Psychosom. Res.* 153, DOI: 10.1016/j.jpsychores.2021.110709 (2022).
- 14. Gao J, Zheng P, Jia Y *et al.* Mental health problems and social media exposure during COVID-19 outbreak. *PLOS ONE* 15(4), e0231924 (2020).
- 15. Suratana S, Tamornpark R, Apidechkul T et al. Impacts of and survival adaptations to the COVID-19 pandemic among the hill tribe population of northern Thailand: a qualitative study. PLOS ONE 16(6), e0252326 (2021).
- Lo A, Jun BO, Shieh V, Starlin CM, Sugahara S, Li DJ. A Comparative Study of Social Impacts of the COVID-19 Pandemic on Republic of Korea, Japan, and Taiwan. *Psych. Investig.* 18(10), 1006–1017 (2021).
- This study showed the societal influential caused of COVID-19 in three East Asia countries.
- 17. Janiri D, Carfi A, Kotzalidis GD *et al.* Posttraumatic Stress Disorder in Patients After Severe COVID-19 Infection. *JAMA Psych.* 78(5), 567–569 (2021).
- Kim JK, Crimmins EM. How does age affect personal and social reactions to COVID-19: results from the national Understanding America Study. PLoS ONE 15(11), e0241950 (2020).
- 19. Luo Y, Chua CR, Xiong Z, Ho RC, Ho CSH. A systematic review of the impact of viral respiratory epidemics on mental health: an implication on the Coronavirus Disease 2019 Pandemic. *Front. Psych.* 11, 565098 (2020).
- 20. Wang C, Tee M, Roy AE *et al.* The impact of COVID-19 pandemic on physical and mental health of Asians: a study of seven middle-income countries in Asia. *PLOS ONE* 16(2), e0246824 (2021).
- 21. Wang C, Chudzicka-Czupała A, Tee ML *et al.* A chain mediation model on COVID-19 symptoms and mental health outcomes in Americans, Asians and Europeans. *Sci. Rep.* 11(1), 6481 (2021).
- 22. Andersen AJ, Mary-Krause M, Bustamante JJH, Héron M, El Aarbaoui T, Melchior M. Symptoms of anxiety/depression during the COVID-19 pandemic and associated lockdown in the community: longitudinal data from the TEMPO cohort in France. *BMC Psych.* 21(1), 381 (2021).
- 23. Ceban F, Nogo D, Carvalho IP et al. Association between mood disorders and risk of COVID-19 infection, hospitalization, and death: a systematic review and meta-analysis. JAMA Psych. 78(10), 1079–1091 (2021).
- 24. Davis HE, McCorkell L, Vogel JM, Topol EJ. Long COVID: major findings, mechanisms and recommendations. *Nat. Rev. Microbiol.* 21(3), 133–146 (2023).
- 25. Renaud-Charest O, Lui LMW, Eskander S *et al.* Onset and frequency of depression in post-COVID-19 syndrome: a systematic review. *J. Psychiatr. Res.* 144, 129–137 (2021).
- 26. Ceban F, Ling S, Lui LMW *et al.* Fatigue and cognitive impairment in Post-COVID-19 Syndrome: a systematic review and meta-analysis. *Brain Behav. Immun.* 101, 93–135 (2022).
- 27. de Miguel-Arribas A, Aleta A, Moreno Y. Impact of vaccine hesitancy on secondary COVID-19 outbreaks in the US: an age-structured SIR model. *BMC Infect. Diseas.* 22(1), 511 (2022).
- d'Arqom A, Asa P, Andriani A *et al.* COVID-19 vaccines knowledge and acceptance among Indonesian adults in Java Island [version 1; peer review: awaiting peer review]. *F1000Research* 12(170), 1–21 (2023).
- 29. Lee Y, Lui LMW, Chen-Li D *et al.* Government response moderates the mental health impact of COVID-19: a systematic review and meta-analysis of depression outcomes across countries. *J. Affect. Disord.* 290, 364–377 (2021).

- Chaudhuri K, Howley P. The impact of COVID-19 vaccination for mental well-being. *European Economic Rev.* 150, DOI: 10.1016/j.euroecorev.2022.104293 (2022).
- 31. Li D-J, Chou L-S, Chou FH-C *et al.* COVID-related psychological distress fully mediates the association from social impact to sleep disturbance among patients with chronic schizophrenia. *Scient. Rep.* 11(1), DOI: 10.1038/s41598-021-96022-2 (2021).
- •• This study showed that the COVID-19 pandemic also affect the schizophrenia patient.
- Li DJ, Kao WT, Shieh V, Chou FH, Lo HA. Development and Implementation of Societal Influences Survey Questionnaire (SISQ) for Peoples during COVID-19 Pandemic: A Validity and Reliability Analysis. Int. J. Environ. Res. Public Health 17(17), 6246 (2020).
- The original article whos develops SISQ, that used in this study.
- 33. Worldometer. Indonesia Population. https://www.worldometers.info/world-population/indonesia-population/ (2023).
- Handayani W, Kumalasari NR. Migration as Future Adaptive Capacity: The Case of Java Indonesia. In: *Environmental Change, Adaptation and Migration: Bringing in the Region.* Hillmann F, Pahl M, Rafflenbeul B, Sterly H (Eds). Palgrave Macmillan, UK London, 117–138 (2015).
- 35. Mulyanto J, Kunst AE, Kringos DS. Geographical inequalities in healthcare utilisation and the contribution of compositional factors: a multilevel analysis of 497 districts in Indonesia. *Health & Place* 60, DOI: 10.1016/j.healthplace.2019.102236 (2019).
- 36. Antara. COVID-19 in Indonesia. https://www.antaranews.com/covid-19 (2023).
- 37. Clemente-Suárez VJ, Navarro-Jiménez E, Moreno-Luna L *et al.* The Impact of the COVID-19 Pandemic on Social, Health, and Economy. *Sustainability* 13(11), 6314 (2021).
- Thiel F, Büechl VCS, Rehberg F et al. Changes in Prevalence and Severity of Domestic Violence During the COVID-19 Pandemic: A Systematic Review. Front. Psychiatry 13, DOI: 10.3389/fpsyt.2022.874183 (2022).
- Quílez-Robres A, Lozano-Blasco R, Íñiguez-Berrozpe T, Cortés-Pascual A. Social, family, and educational impacts on anxiety and cognitive empathy derived from the COVID-19: study on families with children. *Front. Psychol.* 12, DOI: 10.3389/fpsyg.2021.562800 (2021).
- •• This study showed several aspects that affect the mental health of young adults.
- 40. Adams EL, Smith D, Caccavale LJ, Bean MK. Parents are stressed! patterns of parent stress across COVID-19. *Front.Psych.* 12, 626456 (2021).
- 41. Cyr S, Marcil MJ, Houchi C *et al.* Evolution of burnout and psychological distress in healthcare workers during the COVID-19 pandemic: a 1-year observational study. *BMC Psych.* 22(1), 809 (2022).
- 42. Padmanabhanunni A, Pretorius TB. Teacher Burnout in the Time of COVID-19: Antecedents and Psychological Consequences. Int. J. Environ. Res. Public Health 20(5) (2023).
- 43. Sun Y, Zhu S, Chenhuang G et al. COVID-19 burnout, resilience, and psychological distress among Chinese college students. Front. Public Health 10 (2022).
- 44. Joaquim RM, Pinto ALCB, Guatimosim RF et al. Bereavement and psychological distress during COVID-19 pandemics: the impact of death experience on mental health. Curr. Res. Behav. Sci. 2, DOI: 10.1016/j.crbeha.2021.100019 (2021).
- 45. WHO. 7th COVID-19 Intra-Action Review: embracing lessons learned from the COVID-19 response towards health transformation. https://www.who.int/indonesia/news/detail/21-12-2022-7th-covid-19-intra-action-review--embracing-lessons-learned-from-the-covid-19-response-towards-health-transformation (2023).
- Koltai J, Raifman J, Bor J, McKee M, Stuckler D. COVID-19 Vaccination and Mental Health: A Difference-In-Difference Analysis of the Understanding America Study. Am. J. Prev. Med. 62(5), 679–687 (2022).
- 47. Manchia M, Gathier AW, Yapici-Eser H *et al.* The impact of the prolonged COVID-19 pandemic on stress resilience and mental health: a critical review across waves. *Eur. Neuropsychopharmacol.* 55, 22–83 (2022).
- 48. Li K, Luo K, Zhan X *et al.* Adjusting intervention strategies for mental health of COVID-19 patients: a network analysis based on a survey in Omicron-infected patients. *Front. Public Health* 10, DOI: 10.3389/fpubh.2022.1038296 (2022).
- 49. Williamson RE, Hoeboer CM, Primasari I et al. Symptom networks of COVID-19-related versus other potentially traumatic events in a global sample. J. Anxiety Disord. 84, DOI: 10.1016/j.janxdis.2021.102476 (2021).
- 50. Nishimi K, Neylan TC, Bertenthal D, Dolsen EA, Seal KH, O'Donovan A. Post-traumatic stress disorder and risk for hospitalization and death following COVID-19 infection. *Translat. Psych.* 12(1), 482 (2022).
- 51. Rao TS, Asha MR, Ramesh BN, Rao KS. Understanding nutrition, depression and mental illnesses. Indian J. Psych. 50(2), 77-82 (2008).
- 52. Ross CA, Goode C, Schroeder E. The yield from vitamin D screening among psychiatric inpatients. *General Hospital Psych.* 45, 104 (2017).
- 53. Rucklidge JJ, Afzali MU, Kaplan BJ *et al.* Massacre, earthquake, flood: translational science evidence that the use of micronutrients postdisaster reduces the risk of post-traumatic stress in survivors of disasters. *Internat. Perspecti. Psychol.* 10, 39–54 (2021).
- 54. Wang Y, Zhao Y, Zhang J, Bian J, Zhang R. Detecting associations between dietary supplement intake and sentiments within mental disorder tweets. *Health Informatics J.* 26(2), 803–815 (2020).

- 55. Hagan MJ, Roubinov DR, Cordeiro A, Lisha N, Bush NR. Young children's traumatic stress reactions to the COVID-19 pandemic: the long reach of mothers' adverse childhood experiences. J. Affect. Disord. 318, 130–138 (2022).
- 56. Stupar D, Stevanovic D, Vostanis P *et al.* Posttraumatic stress disorder symptoms among trauma-exposed adolescents from low- and middle-income countries. *Child Adolesc. Psych. Mental Health* 15(1), 26 (2021).
- 57. Sheynin J, Beck KD, Pang KC *et al.* Behaviourally inhibited temperament and female sex, two vulnerability factors for anxiety disorders, facilitate conditioned avoidance (also) in humans. *Behav Processes* 103, 228–235 (2014).
- Vilaplana-Pérez A, Sidorchuk A, Pérez-Vigil A *et al.* Assessment of posttraumatic stress disorder and educational achievement in Sweden. JAMA Network Open 3(12), e2028477–e2028477 (2020).
- 59. Tang B, Deng Q, Glik D, Dong J, Zhang L. A Meta-Analysis of Risk Factors for Post-Traumatic Stress Disorder (PTSD) in Adults and Children after Earthquakes. *Int. J. Environ. Res. Public Health* 14(12), 1537 (2017).
- 60. Rumain B, Schneiderman M, Geliebter A. Prevalence of COVID-19 in adolescents and youth compared with older adults in states experiencing surges. *PLoS ONE* 16(3), e0242587 (2021).
- 61. Tan ST, Kwan AT, Rodríguez-Barraquer I *et al.* Infectiousness of SARS-CoV-2 breakthrough infections and reinfections during the Omicron wave. *Nat. Med.* 29(2), 358–365 (2023).
- WHO. COVID-19 disrupting mental health services in most countries, WHO survey. https://www.who.int/news/item/05-10-2020-covid-19-disrupting-mental-health-services-in-most-countries-who-survey (23 July 2023).
- 63. Dondé C, Pouchon A, Pelluet A, Bougerol T, Polosan M. The Impact of Wearing a Face Mask on the Psychiatric Interview: a National Survey During the COVID-19 Pandemic. *Psychiatr. Q.* 93(2), 435–442 (2022).
- 64. Gratzer D, Khalid-Khan F. Internet-delivered cognitive behavioural therapy in the treatment of psychiatric illness. *CMAJ* 188(4), 263–272 (2016).
- 65. Hedman-Lagerlöf E, Carlbring P, Svärdman F, Riper H, Cuijpers P, Andersson G. Therapist-supported Internet-based cognitive behaviour therapy yields similar effects as face-to-face therapy for psychiatric and somatic disorders: an updated systematic review and meta-analysis. *World Psych.* 22(2), 305–314 (2023).
- Effectivity of internet based cognitive behavioural therapy in psychiatric.
- 66. Mahoney A, Li I, Grierson A, Millard M, Haskelberg H, Mason E. Internet-based cognitive behaviour therapy for insomnia before and during the COVID-19 pandemic. *Australian Psychol.* 57(1), 65–76 (2022).
- 67. Young DKW, Carlbring P, Ng PYN, Chen Q-RJ. Feasibility of Self-Guided Online Cognitive Behavioral Therapy for University Students During COVID-19. *Res. Social Work Practice* 32(8), 898–911 (2022).
- 68. You JHS, Luk SWC, Chow DYW, Jiang X, Mak ADP, Mak WWS. Cost-effectiveness of internet-supported cognitive behavioral therapy for university students with anxiety symptoms: A Markov-model analysis. *PLoS ONE* 17(5), e0268061 (2022).
- 69. d'Arqom A, Sawitri B, Nasution Z *et al.* Development of online mental health supporting group to reduce mental burden during covid-19 pandemic. *J. Public Service* 4(2), 251–258 (2020).
- Ifdil I, Fadli RP, Suranata K, Zola N, Ardi Z. Online mental health services in Indonesia during the COVID-19 outbreak. Asian J. Psychiatr. 51, DOI: 10.1016/j.ajp.2020.102153 (2020).
- 71. Geraldina AM, Suen MW, Suanrueang P. Online mental health services during COVID-19 pandemic in Indonesia: challenges from psychologist perspective. *PLOS ONE* 18(6), e0285490 (2023).